Author’s response to reviews

Title: Haloperidol affects bones while clozapine alters metabolic parameters - sex specific effects in rats perinatally treated with phencyclidine

Authors:

Tatjana Nikolic (tatjana.nikolic@med.bg.ac.rs)

Milan Petronijevic (milanpetronijevic@yahoo.com)

Jelena Sopta (jslabic@yahoo.com)

Milica Velimirovic (drmilicavelimirovic@gmail.com)

Tihomir Stojkovic (stihomir85@yahoo.com)

Gordana Jevtic Dozudic (jevtic.gordana86@gmail.com)

Milan Aksic (milan.aksic@med.bg.ac.rs)

Nevena Radonjic (nevence@yahoo.com)

Natasa Petronijevic (natasapetronijevic@yahoo.com)

Version: 2 Date: 20 Sep 2017

Author’s response to reviews:

Dear Editor,

Thank you very much for your constructive comments concerning our manuscript. Those comments were very helpful for improving our paper. We have made the necessary corrections which we hope meet with approval. Herein are answers to specific points:

Editor Comments:

Please include intra- and inter-assay CVs for your ELISA assays.

Response: Thank you for your careful review. We have added intra- and inter-assay CVs for ELISA assays (Methods section, page 10, lines 1-9) according to your suggestion.